Author's response to reviews

Title: Protective effects of stem bark of Harungana madagascariensis on the red blood cell membrane

Authors:

Prosper C N Biapa (prbiapa@yahoo.fr)
Horea Matei (hmatei@umfcluj.ro)
Ştefana Bâlici (stefanabalici@yahoo.com)
Julius E Oben (juliusoben@hotmail.com)
Jeanne Y Ngogang (jngogang@yahoo.fr)

Version: 6 Date: 4 March 2013

Author's response to reviews: see over
Dear Mr. James Ellis D. Prozenko
Journal Editorial Office
BioMedCentral

Please find attached the latest form of our manuscript entitled:
Protective effects of stem bark of Harungana madagascariensis on the red blood cell membrane
(Prosper C N Biapa, Horea Matei, Ştefana Bâlici, Julius E Oben and Jeanne Y Ngogang)

Concerning the most recent report of Reviewer 2 (Patcharee Boonsiri), the answers for his comments are as follows:

The authors must intensively check and correct results and discussion parts. The way of presentation Figure and Table is important for the reader to understand your work.

Answer:
Discussion and Results were revised in the manuscript uploaded.
Figures and Tables were modified according to reviewers’ requests.

Quality of written English: Needs some language corrections before being published.

Answer:
Done

We hope this version will satisfy the reviewer’s requests.

Sincerely yours,
Horea Matei, MD, PhD
Head of the Cell and Molecular Biology Department
“Iuliu Hațieganu” University of Medicine and Pharmacy
Cluj-Napoca, Romania